Author's response to reviews

Title: Observational Skills Assessment Score: reliability in measuring amount and quality of use of the affected hand in unilateral Cerebral Palsy

Authors:

Lucianne A Speth (l.speth@adelante-zorggroep.nl)
Yvonne J Janssen-Potten (Y.Janssen@adelante-zorggroep.nl)
Pieter Leffers (p.leffers@maastrichtuniversity.nl)
Eugene A Rameckers (eaa.rameckers@hetnet.nl)
Anke Defesche (a.defesche@adelante-zorggroep.nl)
Richard P Geers (r.geers@adelante-zorggroep.nl)
Rob J Smeets (rob.smeets@mumc.nl)
Hans J.S.H. Vles (jsh.vles@mumc.nl)

Version: 2 Date: 29 January 2013

Author's response to reviews:

I forgot the authors' contributions:

LS is first corresponding author and participated in drafting the manuscript. She made main contributions to design, acquisition, analysis and interpretation of data. YJ and PL have been involved in drafting the manuscript, analysis and interpretation of data and critically revising it. ER and AD were involved in data acquisition and development of the OSAS. RG designed the user-dedicated software program of the OSAS. RS and HV critically revised the intellectual content of the article. All authors read and approved the final manuscript.